Author's response to reviews

Title: TOX3 is Expressed in Mammary ER+ Epithelial Cells and Regulates ER Target Genes in Luminal Breast Cancer

Authors:

Akop Seksenyan (Akop.Seksenyan@my.rosalindfranklin.edu)
Asha Kadavallore (Kadavallorea@csmc.edu)
Ann E Walts (Ann.Walts@cshs.org)
Brian de la Torre (Delatorreb@csmc.edu)
Dror Berel (berel@usc.edu)
Samuel P Strom (SStrom@mednet.ucla.edu)
Parinaz Aliahmad (Aliahmadp@csmc.edu)
Vincent Funari (FunariV@cshs.org)
Jonathan Kaye (kayej@csmc.edu)

Version: 2 Date: 2 September 2014

Author's response to reviews:

This revised version has been modified as follows:

1. Inclusion of revised statements regarding use of animals and normal human breast tissue in Methods.

2. As indicated in the original submission, microarray data was deposited in GEO and the accession number was given. This information has now been moved to a new subsection ‘Availability of Supporting Data’.